

Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL078691

- Simulated impacts of the NAO on the interannual variability of precipitation in summer in eastern China are investigated
- The model CCSM 2.0.1 was able to of the teleconnection climate
- Precipitation anomalies in eastern China associated with the summer NAO are linked to the "North Atlantic-Eurasian teleconnection pattern'

Supporting Information:

Supporting Information S1

Correspondence to: Y. Pena voubingpeng@mail.xitu.edu.cn

Citation:

Peng, Y. (2018). Simulated interannual teleconnection between the summer North Atlantic Oscillation and summer precipitation in eastern China during the last millennium. Geophysical Research Letters, 45, 7741-7747. https:// doi.org/10.1029/2018GL078691

Received 29 MAY 2018 Accepted 19 JUL 2018 Accepted article online 25 JUL 2018 Published online 11 AUG 2018

Kev Points:

- roughly produce the general features influences in eastern China

©2018. American Geophysical Union. All Rights Reserved.

Simulated Interannual Teleconnection Between the Summer North Atlantic Oscillation and Summer Precipitation in Eastern China During the Last Millennium

Youbing Peng¹

¹School of Human Settlements and Civil Engineering, Xi'an Jiaotong University, Xi'an, China

Abstract The interannual teleconnections between the summer North Atlantic Oscillation and summer precipitation in eastern China over the last millennium are investigated in a 1,000-year simulation by the global climate model Community Climate System Model version 2.0.1. The model was able to roughly produce the general features of the teleconnection influences in eastern China, with negative (positive) precipitation anomalies north of 30°N and positive (negative) precipitation anomalies south of 30°N, associated with the positive (negative) phase of the North Atlantic Oscillation. The composite analysis reveals that the influence of the North Atlantic Oscillation on precipitation in summer in eastern China is chiefly via the "North Atlantic-Eurasian teleconnection pattern," which extends from the subtropical North Atlantic to East Asia.

Plain Language Summary The impacts of the North Atlantic Oscillation on the climate in eastern China have been highlighted over the last decade. Recently, the observed impacts of the North Atlantic Oscillation on precipitation in eastern China in summer, chiefly via the "North Atlantic-Eurasian teleconnection pattern," have been noted. This pattern propagates along a high-latitude pathway and is different from the midlatitude circumglobal teleconnection pattern. Here this teleconnection pattern is investigated in a millennium context using a 1,000-year simulation of the Community Climate System Model version 2.0.1. The model results reveal that the simulated NAO-related teleconnection pattern during the summer season is very similar to the observed pattern. Along its path through Eurasia, a positive NAO index causes anomalously high sea level pressures and 500-hPa geopotential heights over Mongolia-north China that correspond to the anomalous anticyclone in the 850-hPa wind anomalies and hence cause reduced precipitation in north China. The opposite occurs during negative summer NAO phases. The model's teleconnection patterns of the summer NAO capture the reconstructed and observed patterns of the summer precipitation changes in eastern China and provide further evidence for the climate model-based implications for predictions and future projections of seasonal precipitation in eastern China.

1. Introduction

The North Atlantic Oscillation (NAO) is a primary mode of internal dynamics in the atmosphere and is characterized by the difference in sea level pressure (SLP) between the Azores high and Icelandic low (Hurrell, 1995). The NAO governs large changes in surface temperature, precipitation, cloudiness, etc., around the North Atlantic region (Hurrell et al., 2003). However, the NAO also has a downstream climate influence over remote regions via the NAO-related downstream wave train. For example, remote influences of the NAO on the interannual variability in summer precipitation in eastern China have been explored to some extent. Early studies on this subject have focused on the influence of the winter or spring NAO (Bollasina & Messori, 2018; Fu & Zeng, 2005; Gu et al., 2009; Sung et al., 2006; Tian & Fan, 2012; Wu et al., 2009, 2012; Yang et al., 2004; Zheng et al., 2016; Zuo et al., 2012) and have suggested that their influence on East Asian summer precipitation is chiefly via the subtropical Asian jet waveguide-trapped wave train (e.g., Bollasina & Messori, 2018; Sung et al., 2006). Recently, a high-latitude remote climatic teleconnection in summer in eastern China from the North Atlantic Ocean has garnered considerable attention. Wu et al. (2009, 2012) and Zuo et al. (2012) investigated the influences of the spring NAO on the East Asian summer monsoon. They found that spring NAO can trigger a tripole pattern of sea surface temperature anomalies (SSTAs) in the North Atlantic Ocean. The tripole SSTA pattern can persist into the following summer and excite a remote teleconnection over northern Eurasia. This teleconnection pattern, which originates in the subtropical North Atlantic, is different from the pattern with the Asian jet waveguide path and was first termed the "North Atlantic-Eurasian teleconnection pattern" by Li and Ruan (2018). The summer NAO variation has been shown to influence large-scale summer precipitation anomalies over East Asia via this teleconnection pattern (Bollasina & Messori, 2018; Folland et al., 2009; Linderholm et al., 2011, 2013; Liu & Yin, 2001; Zheng et al., 2016). Significant negative correlations between the teleconnection pattern index and summer precipitation were found in northern China. However, most of these previous studies have focused on interannual teleconnections between the NAO and precipitation in summer over eastern China during the instrumental period, with relatively less emphasis on this topic beyond the era of instrumental measurements.

An increasing number of high-resolution proxy records and global climate model simulations covering the last millennium make it possible to obtain more insight into this teleconnection on long, century-length time scales. Several studies using tree ring records extended summer NAO-related hydroclimate changes in eastern China several centuries back in time (Folland et al., 2009; Linderholm et al., 2013; Linderholm & Folland, 2017). These studies suggested that the influence of the teleconnection pattern related to the summer NAO on precipitation in eastern China could also be found in the climate change beyond the instrumental period (Folland et al., 2009; Linderholm et al., 2013). However, few studies have focused on determining whether the model is simulating the interannual teleconnections between the summer NAO and summer precipitation in eastern China as revealed by the observations and reconstructions. The objective of this study is to examine the relationship between the summer NAO and eastern China's summer precipitation during the last millennium based on model data from a 1,000-year global climate model simulation.

2. Data and Methods

To investigate the regional summer precipitation changes in eastern China associated with the summer NAO, we analyzed a 1,000-year (1,000-1,999) simulation performed by the Community Climate System Model (CCSM, version 2.0.1). The CCSM was developed by the National Center for Atmospheric Research (NCAR; Kiehl & Gent, 2004). The main forcings driving the model include global and seasonal changes in orbital insolation (Berger, 1978), solar variations and volcanic eruptions (Crowley et al., 2003), and greenhouse gases (Ammann et al., 2007). A model-observation comparison shows that the model effectively simulates the zonal gradient of summer precipitation over East Asia, which is very similar to the observed pattern (Shen et al., 2009). A model-proxy comparison shows the simulation produces a temporal variability similar to that observed in proxy data in eastern China during the last millennium (Peng et al., 2009, 2014). The modeled summer NAO index (NAOI; shown in Figure S1) was defined as the June-July-August mean of the difference between normalized SLP anomalies over two boxes in the North Atlantic, a low-latitude/high-pressure box located between 20°N-40°N and 70°W-10°W and a high-latitude/low-pressure box located between 45°N-65°N and 50°W-10°E (Lorenz et al., 2006). The definition of the NAOI here is based on the permanent action centers of the Azores high and Icelandic low because the two boxes are located at the same latitudes as the Azores high and Icelandic low, respectively, as suggested by the observation (Hameed & Piontkovski, 2004). Through a systematic comparison of the NAO indices, Pokorná and Huth (2015) suggested that both the central-based NAO indices and the principal component analysis-based NAO indices are equally physically plausible. However, the action centers of high covariability defined by the principal component analysis show a northward migration from the Azores high and Icelandic low in summer (Pokorná & Huth, 2015).

To further investigate whether the characteristics of the interannual teleconnection between the NAO and precipitation in summer in eastern China revealed by CCSM2.0.1 are robust across multiple model experiments, monthly mean precipitation flux data from the past 1,000 simulations of the Coupled Model Intercomparison Project phase 5 (CMIP5) and the Paleoclimate Model Intercomparison Project phase 3 (PMIP3) are also used here. Atmosphere-ocean general circulation models from the CMIP5/PMIP3 used in our study include CCSM4, CSIRO-Mk3L-1-2 (CSIRO), MPI-ESM-P (MPI), and HadCM3.

Additionally, the availability of the tree ring-based Monsoon Asia Drought Atlas (MADA; Cook et al., 2010) and a reconstructed NAOI developed by Ortega et al. (2015) are used here to explore remote influences from the NAO on the interannual variability in summer precipitation in eastern China during the last millennium. The data underlying the MADA provide annual tree ring drought reconstruction for 1300–2005 AD on a 2.5° by 2.5° gridded network of summer (June-July-August) Palmer drought severity index (PDSI) data. The proxy record of NAOI is available for 1049–1969 AD at an annual resolution.

Geophysical Research Letters



Figure 1. (a and b) Composites of the JJA hydroclimate patterns for the positive and negative NAOI calculated using simulated data for 1000–1999 and (c and d) the reconstructions by Cook et al. (2010) and Ortega et al. (2015) for 1300–1969. The unit of the summer precipitation is mm day⁻¹. The crosses denote the five action centers of the "North Atlantic-Eurasian teleconnection pattern." Positive (negative) NAOI is defined as NAOI above (below) 1 (-1) standard deviation. SS on the top right of each plot indicates the sample size. Anomalies with statistical significance levels higher than 90% are marked with black dots.

The relationship between NAO and precipitation changes over eastern Asia is evaluated by composite analysis. The significance of the composite analysis is evaluated using a two-sample Student's *t* test that is assumed to have different population variances.

3. Results

The simulated relationship between NAO and precipitation changes over eastern Asia is evaluated by compositing the summer precipitation for positive or negative NAOI (Figures 1a and 1b). The 1,000-year simulated data suggest that the teleconnected precipitation pattern in eastern China with the NAO is characterized by a north-south dipole pattern, with negative (positive) precipitation anomalies located north of 30°N and positive (negative) precipitation anomalies located south of 30°N associated with the positive (negative) anomalies of the NAOI; however, the relationship in southern China is not statistically significant. The same type of analysis was applied to the MADA PDSI and shows that there is close agreement between the NAO-related MADA PDSI patterns and simulated hydroclimate impacts in eastern China. The impact of the positive (negative) NAO phases in eastern China revealed by MADA is also "dipolar," with the negative (positive) center located north of 35°N and the positive (negative) center located south of 35°N (Figures 1c and 1d). However, there is an uncertainty about whether the PDSI composites shown in Figures 1c and 1d are related to the winter or summer NAO in the proxy data with an annual resolution. These climatic effects of the teleconnection pattern in eastern China revealed by the simulation and MADA PDSI are very similar to the observations, as shown by Figure 5b in Li and Ruan (2018). Their observation results indicate that such precipitation anomalies in eastern China associated with the NAO are strongly linked to a high-latitude Rossby wave train pattern. This train pattern, termed the "North Atlantic-Eurasian teleconnection pattern," has five centers of action from the subtropical North Atlantic Ocean across the Eurasian continent. This five-pole pattern extending from the North Atlantic to East Asia could be observed in Figures 1a and 1b and is similar to the observations, as shown by Li and Ruan (2018). Under positive NAOI in summer, less precipitation occurs in the subtropical North Atlantic Ocean (A in Figure 1a), Eastern Europe (C in Figure 1a), and Mongolia-north China (E in Figure 1a), and more precipitation occurs in the northeastern North Atlantic Ocean (B in

Geophysical Research Letters



Figure 2. (a and b) Composites of the JJA sea surface pressure (contour interval is 20 Pa) and (c and d) geopotential height at 500 hPa (contour interval is 2.5 gpm) for the positive and negative NAOI calculated using simulated data for the last millennium. Positive (negative) NAOI is defined as NAOI above (below) 1 (-1) standard deviation. SS on the top right of each plot indicates the sample size. Anomalies with statistical significance levels higher than 90% are marked with black dots.

Figure 1a) and Urals (D in Figure 1a). During negative NAOI, the patterns are largely reversed (Figure 1b). The five-pole pattern observed in Figures 1a and 1b indicates that the interannual teleconnection pattern between the simulated summer NAO and precipitation in eastern China may be the eastern end of the regional climate response linked to the wave train along its path through northern Eurasia.

To check if the five-pole pattern is physically consistent with the summer precipitation composite results, the simulated SLP anomalies and geopotential height anomalies at 500 hPa composited using the simulated NAOI were examined (Figure 2). The most striking feature identified in the SLP anomalies is a five-core wave train pattern extending from the subtropical North Atlantic to East Asia (Figures 2a and 2b), similar to the precipitation anomaly pattern in Figures 1a and 1b. The spatial pattern of the first two centers located over the North Atlantic bears the structure of the NAO as expected. The patterns seen in the geopotential height anomalies at 500 hPa are qualitatively similar to those found in the SLP composite analysis, except that they are stronger over Eastern Europe (Figures 2c and 2d). The composite anomalies in the geopotential height anomalies at 500 hPa that are associated with the NAOI indicate that three positive (negative) geopotential height anomaly regions occur over the subtropical North Atlantic, Eastern Europe, and Mongolia-north China, and another two negative (positive) geopotential height anomaly regions occur over the high-latitude North Atlantic and the Ural Mountains during positive (negative) NAOI years. This five-core Rossby wave train associated with the NAO has been discussed by Li and Ruan (2018) and Li et al. (2008) during the instrumental period. They suggested that the five-core Rossby wave train should be regarded as another teleconnection pattern between the North Atlantic Ocean and the Eurasian continent because it generally propagates along a high-latitude pathway, which is different from the midlatitude circumglobal teleconnection pattern. As shown in Figure 3, the positive anomaly centers in the pressure field of the wave train correspond to the centers of the anomalous anticyclone in the 850-hPa wind anomalies and vice versa. The anomalous anticyclonic circulation centers linked with the wave train originating from the North Atlantic tend to accompany anomalous subsidence and reduced precipitation in the regions where they are located. In contrast, surplus





Figure 3. Same as Figure 2 but for the composites of JJA wind (m s⁻¹) at 850 hPa for the (a) positive and (b) negative NAOI. The wind vectors are missing over western China and around where the simulated 850-hPa surface is below the ground.

precipitation due to the anomalous ascent of air is dominant over the anomalous cyclonic circulation locations. Thus, we conclude that the NAO variation contributes to the simulated anomalous summer precipitation patterns, as shown in Figures 1a and 1b, via the five-core wave train pattern, as shown in Figure 2.

4. Conclusions and Discussion

We use a 1,000-year simulation performed with the CCSM2.0.1 to characterize the interannual teleconnections between the summer NAO and summer precipitation in eastern China during the last millennium. The result of the composite analysis reveals that the positive (negative) NAOI is associated with negative (positive) precipitation anomalies at approximately north of 30°N and positive (negative) precipitation south of this latitude. These NAO impacts over eastern China agree with the MADA and observation data. The influence of NAO on summer precipitation in eastern China is chiefly via the "North Atlantic-Eurasian teleconnection pattern." A positive NAOI is indicative of anomalously high SLP and 500-hPa geopotential heights over the subtropical North Atlantic Ocean, Eastern Europe, and Mongolia-north China and low SLP and geopotential heights over the high-latitude North Atlantic Ocean and Urals, and vice versa. The positive anomaly centers in the pressure field of the wave train correspond to the centers of the anomalous anticyclone in the 850-hPa wind anomalies and hence causes deficient precipitation in the regions where they are located and vice versa. Thus, during the positive phase of NAOI, deficient precipitation regions occur over the subtropical North Atlantic Ocean, Eastern Europe, and Mongolia-north China, and positive precipitation anomaly regions occur in the northeastern North Atlantic Ocean and Urals. In each of these regions, the precipitation anomalies exhibit opposite signs associated with the negative NAOI. Our simulated results show that the interannual teleconnection pattern between the simulated summer NAO and precipitation in eastern China may be the eastern end of the regional climate response linked to the wave train along its pathway through northern Eurasia. However, the questions that arise are whether the teleconnection influences and mechanism due to the NAO in summer over the last millennium as a whole revealed by CCSM2.0.1 are robust during those periods with distinct climatic characteristics and across different model experiments? Answering these questions may lead to a better understanding of the interannual teleconnection between the summer NAO and summer precipitation in eastern China during the last millennium.

There are three distinct climate intervals during the last millennium: the Medieval Climate Anomaly (AD 1000–1300), the Little Ice Age (AD 1300–1850), and the Current Warm Period (AD 1850 to present). The changes in climate during the Medieval Climate Anomaly and Little Ice Age are controlled mainly by natural forcing, while anthropogenic factors play a large role in the climate changes during the Current Warm Period. A comparison of the interannual teleconnection between the summer NAO and summer precipitation in eastern China among the three climate intervals shows that the teleconnection influences and mechanism during those periods are all similar to those during the last millennium as a whole (Figures S2 and S3). This comparison indicates that the simulated results identified a robust climatic impact due to the NAO in summer during the last millennium.

The results of a model intercomparison based on four simulations from the CMIP5/PMIP3 archive show that the five-pole precipitation anomaly pattern due to the NAO in summer revealed by CCSM2.0.1 can also be observed in different model experiments, except for CCSM4 (Figures S4 and S5); this finding supports the conclusion that the influence of the NAO on summer precipitation over northern Eurasia is chiefly via the "North Atlantic-Eurasian teleconnection pattern." However, the models differ in their teleconnection influences in eastern China, and the causes of these differences among these atmosphere-ocean general circulation models is another associated topic that remains to be explored.

References

- Ammann, C., Joos, F., Schimel, D., Otto-Bliesner, B., & Tomas, R. (2007). Solar influence on climate during the past millennium: Results from transient simulations with the NCAR climate system model. *Proceedings of the National Academy of Sciences of the United States of America*, 104(10), 3713–3718. https://doi.org/10.1073/pnas.0605064103
- Berger, A. (1978). Long-term variations of daily insolation and quaternary climatic changes. *Journal of the Atmosphereic Sciences*, 35(12), 2362–2367. https://doi.org/10.1175/1520-0469(1978)035<2362:LTVODI>2.0.CO;2
- Bollasina, M. A., & Messori, G. (2018). On the link between the subseasonal evolution of the North Atlantic Oscillation and east Asian climate. *Climate Dynamics*, 67, 1–21. https://doi.org/10.1007/s00382-018-4095-5
- Cook, E. R., Anchukaitis, K. J., Buckley, B. M., D'Arrigo, R. D., & Jacoby, G. C. (2010). Asian monsoon failure and megadrought during the last millennium. *Science*, 328(5977), 486–489. https://doi.org/10.1126/science.1185188
- Crowley, T., Baum, S., Kim, K., Hegerl, G., & Hyde, W. (2003). Modelling ocean heat content changes during the last millennium. *Geophysical Research Letters*, 30(18), 1932. https://doi.org/10.1029/2003GL017801
 - Folland, C. K., Knight, J., Linderholm, H. W., Fereday, D., Ineson, S., & Hurrell, J. W. (2009). The summer North Atlantic Oscillation: Past, present, and future. *Journal of Climate*, 22(5), 1082–1103. https://doi.org/10.1175/2008JCLI2459.1

Fu, C., & Zeng, Z. (2005). Correlations between North Atlantic Oscillation index in winter and eastern China flood/drought index in summer in the last 530 years. *Chinese Science Bulletin*, 50(21), 2505–2516. https://doi.org/10.1007/BF03183642

Gu, W., Li, C. Y., Li, W. J., Zhou, W., & Chan, J. C. L. (2009). Interdecadal unstationary relationship between NAO and East China's summer precipitation patterns. *Geophysical Research Letters*, *36*, L13702. https://doi.org/10.1029/2009GL038843

Hameed, S., & Piontkovski, S. (2004). The dominant influence of the Icelandic low on the position of the Gulf stream northwall. *Journal of Geophysical Research*, *31*, L09303. https://doi.org/10.1029/2004GL019561

Hurrell, J. W. (1995). Decadal trends in the North Atlantic Oscillation: Regional temperatures and precipitation. Science, 269(5224), 676–679. https://doi.org/10.1126/science.269.5224.676

Kiehl, J., & Gent, P. (2004). The community climate system model, version 2. *Journal of Climate*, 17(19), 3666–3682. https://doi.org/10.1175/ 1520-0442(2004)017<3666:TCCSMV>2.0.CO;2

Acknowledgments

This work was supported by the National Natural Science Foundation of China (grant 41605046). We acknowledge the critical comments from the anonymous reviewers and editor. The tree ring-based Monsoon Asia Drought Atlas and reconstructed NAOI used in Figure were obtained from NOAA at http://www.ncdc.noaa.gov/data-access/ paleoclimatology-data/datasets/climate-reconstruction. Monthly mean precipitation flux data from the past 1000 experiments of the CMIP5/PMIP3 are available at http://esgf-node.llnl. gov/projects/cmip5.

Hurrell, J. W., Kushnir, Y., Ottersen, G., & Visbeck, M. (2003). An overview of the North Atlantic Oscillation. AGU Geophysical Monograph, 134, 1–35. https://doi.org/10.1029/134GM01

- Li, J., Yu, R. C., & Zhou, T. J. (2008). Teleconnection between NAO and climate downstream of the Tibetan Plateau. Journal of Climate, 21(18), 4680–4690. https://doi.org/10.1175/2008JCLI2053.1
- Li, J. P., & Ruan, C. Q. (2018). The North Atlantic-Eurasian teleconnection in summer and its effects on Eurasian climates. *Environmental Research Letters*, 13(2), 024007. https://doi.org/10.1088/1748-9326/aa9d33
- Linderholm, H. W., & Folland, C. K. (2017). Summer North Atlantic Oscillation (SNAO) variability on decadal to palaeoclimate time scales. Past global changes magazine, 25(1), 57–60.
- Linderholm, H. W., Ou, T. H., Jeong, J. H., Folland, C. K., Gong, D. Y., et al. (2011). Interannual teleconnections between the summer North Atlantic Oscillation and the East Asian summer monsoon. *Journal of Geophysical Research*, *116*, D13107. https://doi.org/10.1029/ 2010JD015235
- Linderholm, H. W., Seim, A., Ou, T., Jeong, J. H., Liu, Y., Wang, X., et al. (2013). Exploring teleconnections between the summer NAO (SNAO) and climate in East Asia over the last four centuries—A tree-ring perspective. *Dendrochronologia*, 31(4), 297–310. https://doi.org/10.1016/j.dendro.2012.08.004
- Liu, X., & Yin, Z. Y. (2001). Spatial and temporal variation of summer precipitation over the eastern Tibetan Plateau and the North Atlantic Oscillation. *Journal of Climate*, 14(13), 2896–2909. https://doi.org/10.1175/1520-0442(2001)014<2896:SATVOS>2.0.CO;2
- Lorenz, S. J., Kim, J. H., Rimbu, N., Schneider, R. R., & Lohmann, G. (2006). Orbitally driven insolation forcing on Holocene climate trends: Evidence from alkenone data and climate modeling. *Paleoceanography and Paleoclimatology*, 21, PA1002. https://doi.org/10.1029/ 2005PA001152
- Ortega, P., Lehner, F., Swingedouw, D., Masson-Delmotte, V., Raible, C., Casado, M., & Yiou, P. (2015). A model-tested North Atlantic Oscillation reconstruction for the past millennium. *Nature*, *523*(7558), 71–74. https://doi.org/10.1038/nature14518

Peng, Y. B., Shen, C. M., Cheng, H., & Xu, Y. (2014). Modeling of severe persistent droughts over eastern China during the last millennium. *Climate of the Past*, 10(3), 1079–1091. https://doi.org/10.5194/cp-10-1079-2014

- Peng, Y. B., Xu, Y., & Jin, L. (2009). Climate changes over eastern China during the last millennium in simulations and reconstructions. *Quaternary International*, 208(1-2), 11–18. https://doi.org/10.1016/j.quaint.2009.02.013
- Pokorná, L., & Huth, R. (2015). Climate impacts of the NAO are sensitive to how the NAO is define. *Theoretical and Applied Climatology*, 119(3-4), 639–652. https://doi.org/10.1007/s00704-014-1116-0
- Shen, C. M., Wang, W. C., Peng, Y. B., Xu, Y., & Zheng, J. Y. (2009). Variability of summer precipitation over eastern China during the last millennium. *Climate of the Past*, 5(2), 129–141. https://doi.org/10.5194/cp-5-129-2009
- Sung, M. K., Kwon, W. T., Baek, H. J., Boo, K. O., Lim, G. H., & Kug, J. S. (2006). A possible impact of the North Atlantic Oscillation on the east Asian summer monsoon precipitation. *Geophysical Research Letters*, 33, L21713. https://doi.org/10.1029/2006GL027253
- Tian, B. Q., & Fan, K. (2012). Relationship between the late spring NAO and summer extreme precipitation frequency in the middle and lower reaches of the Yangtze River. Atmospheric and Oceanic Science Letters, 5(6), 455–460. https://doi.org/10.1080/16742834.2012.11447038
- Wu, Z. W., Li, J. P., Jiang, Z. H., He, J. H., & Zhu, X. (2012). Possible effects of the North Atlantic Oscillation on the strengthening relationship between the east Asian summer monsoon and ENSO. *International Journal of Climatology*, 32(5), 794–800. https://doi.org/10.1002/ joc.2309
- Wu, Z. W., Wang, B., Li, J. P., & Jin, F.-F. (2009). An empirical seasonal prediction model of the east Asian summer monsoon using ENSO and NAO. Geophysical Research Letters, 114, D11107. https://doi.org/10.1029/2008JD011501
- Yang, S., Lau, K. M., Yoo, S. H., Kinter, J. L., Miyakoda, K., & Ho, C. H. (2004). Upstream subtropical signals preceding the Asian summer monsoon circulation. *Journal of Climate*, 17(21), 4213–4229. https://doi.org/10.1175/JCLI3192.1

Zheng, F., Li, J. P., Li, Y. J., Zhao, S., & Deng, D. F. (2016). Influence of the summer NAO on the spring-NAO-based predictability of the east Asian summer monsoon. *Journal of Applied Meteorology and Climatology*, 55(7), 1459–1476. https://doi.org/10.1175/JAMC-D-15-0199.1

Zuo, J. Q., Li, W. J., Ren, H. L., & Chen, L. J. (2012). Change of the relationship between spring NAO and east Asian summer monsoon and its possible mechanism. *Chinese Journal Geophysical*, 55, 384–395.